\begin{table}%t2
\caption{Source list.\label{table:list}}
%\centerline
{\begin{tabular}{lrrccl}
\hline\hline
Source
          & RA (1950)                      & Dec (1950)                   & $D$    &Points  & Associated \\
          & (${\rm (^h)\  (^m)\  (^s)\ }$) &($\degr$ $\arcmin$ $\arcsec$) & (kpc)  &per map & objects    \\
\noalign{\smallskip}\hline\noalign{\smallskip}
RNO~1B        &00 33 53.3   &63 12 32    &  0.85$^a$  & 70  & L1287, G~121.30+0.66, IRAS~00338+6312\\
NGC~281       &00 49 29.2   &56 17 36    &  3.5$^b$  & 34  & S184, G~123.07-6.31 \\
S~187         &01 19 58.0   &61 34 32    &  1.0$^c$  & 67  & G~126.68-0.83       \\
G~133.69+1.22 &02 21 40.8   &61 53 26    &  2.1$^d$  & 52  & W3                  \\
S~199         &02 57 36.9   &60 17 32    &  2.1$^d$  & 46  & IC1848, AFGL~4029   \\
S~201         &02 59 22.4   &60 16 12    &  2.1$^e$  & 41  & AFGL~437, G~138.50+1.64 \\
AFGL~490      &03 23 34.0   &58 33 47    &  0.9$^f$  & 18  \\
G142.00+1.83  &03 23 41.0   &58 36 52    &  0.9$^f$  & 37  & AFGL~490\\
Per~4         &03 26 12.1   &31 17 13    &  0.35$^g$  & 78  \\
LkH$\alpha$101     &04 26 57.3   &35 09 56    &  0.8$^h$  & 9   & AFGL~585, S222\\
G~173.48+2.45 &05 35 51.3   &35 44 16    &  2.3$^d$  & 54  & S231 \\
G~173.58+2.44 &05 36 06.3   &35 39 21    &  2.3$^d$  & 41\\
AFGL~6366     &06 05 40.1   &21 31 16    &  2.0$^i$  & 50  & S247\\
S~255         &06 09 58.2   &18 00 17    &  2.5$^d$  & 67  & AFGL~896\\
LkH$\alpha$25      &06 37 59.5   &09 50 53    &  0.8$^j$  & 1   & AFGL~4519S, NGC~2264, S273\\
S~76E         &18 53 47.0   &07 49 25    &  2.1$^k$  & 61  & G40.50+2.54\\
S~88B         &19 44 41.4   &25 05 17    &  2.0$^d$  & 25\\
DR~21         &20 37 13.5   &42 12 15    &  3.0$^l$  & 56  & W75\\
S~140         &22 17 41.1   &63 03 44    &  0.9$^d$  & 82  & L1204\\
L~1251T4      &22 27 10.2   &74 58 00    &  0.35$^m$  & 119\\
S~153         &22 56 42.0   &58 28 45    &  4.0$^d$  & 42\\
G~264.28$+$1.48  &08 54 39.0    &$-$42 53 30   & 1.4$^n$ & 43\\
G~265.14$+$1.45  &08 57 36.3    &$-$43 33 38   & 1.7$^n$ & 64\\
G~267.94$-$1.06  &08 57 21.7    &$-$47 19 04   & 0.7$^n$ & 9\\
G~268.42$-$0.85  &09 00 12.1    &$-$47 32 07   & 1.3$^n$ & 60\\
G~269.11$-$1.12  &09 01 51.6    &$-$48 16 42   & 2.6$^n$ & 53\\
G~270.26$+$0.83  &09 14 58.0    &$-$47 44 00   & 2.6$^n$ & 27\\
G~285.26$-$0.05  &10 29 36.8    &$-$57 46 40   & 4.7$^n$ & 59\\
G 291.27$-$0.71  &11 09 42.0    &$-$61 01 55   & 2.7$^o$ & 104\\
G~294.97$-$1.73  &11 36 51.6    &$-$63 12 09   & 1.2$^n$ & 39\\
G~305.36$+$0.15  &13 09 21.0    &$-$62 21 43   & 4.2$^n$ & 18\\
G~316.77$-$0.02  &14 41 10.4    &$-$59 35 50   & 3.1$^p$ & 136 \\
G~345.01$+$1.80  &16 53 17.3    &$-$40 09 23   & 2.1$^p$ & 103 & RCW~116\\
G~345.41$-$0.94  &17 06 02.2    &$-$41 32 06   & 2.8$^p$ & 71  & RCW~117\\
G~351.41$+$0.64  &17 17 32.3    &$-$35 44 03   & 1.7$^q$ & 88  & NGC~6334I\\
\hline 
\end{tabular}}
\smallskip
\par
$^a$ Yang et~al. (\cite{yang}),
$^b$ Martin \& Henning (\cite{martin}),
$^c$ Fich \& Blitz (\cite{fich}),
$^d$ Blitz et~al. (\cite{blitz}),
$^e$ assumed,
$^f$ Harvey (\cite{harvey2}),
$^g$ Paper~I,
$^h$ Lada (\cite{lada}),
$^i$ Carpenter et~al. (\cite{carpenter}),
$^j$ Snell et~al. (\cite{snell}),
$^k$ Plume et~al. (\cite{plume}),
$^l$ Harvey et~al. (\cite{harvey1}),
$^m$ Toth \& Walmsley (\cite{toth}),
$^n$ Zinchenko et~al. (\cite{zin3}),
$^o$ Brand \& Blitz (\cite{brand}),
$^p$ Juvela (\cite{juvela}),
$^q$ Neckel (\cite{neckel}).
\end{table}